Author's response to reviews

Title: A RESTful pseudonymization interface for use in modern web applications

Authors:

Martin Lablans (lablans@uni-mainz.de)
Andreas Borg (borga@uni-mainz.de)
Frank Ückert (ueckert@uni-mainz.de)

Version: 2
Date: 26 September 2014

Author's response to reviews:

Dear Arlene Pura,

thank you for your recent email. As per your request, we have revised the document:

- The document is now formatted in a single column.
- Line numbers have been added.
- The correspondence information has been extended to show the email addresses of AB and FU and the full institution/postal address of ML.
- An acknowledgement section has been added.

Also, information about usage of the TMF PID Generator has been updated as requested by the TMF office.

Regards

Martin Lablans